Author’s response to reviews

Title: Withdrawal of antihypertensive therapy in people with dementia: feasibility study

Authors:

Veronika van der Wardt (v.vanderwardt@nottingham.ac.uk)
Jennifer Burton (Jenni.Harrison@ed.ac.uk)
Simon Conroy (spc3@leicester.ac.uk)
Tomas Welsh (tomas.welsh@nhs.net)
Pip Logan (pip.logan@nottingham.ac.uk)
Jaspal Taggar (Jaspal.Taggar@nottingham.ac.uk)
Lukasz Tanajewski (Lukasz.Tanajewski@nottingham.ac.uk)
John Gladman (John.Gladman@nottingham.ac.uk)

Version: 1 Date: 18 Sep 2017

Author’s response to reviews:

Dear Editor,

We would like to thank the reviewers for their comments regarding our manuscript ‘Withdrawal of antihypertensive therapy in people with dementia: feasibility study’. Please find attached the amended version following the reviewers’ comments (comments and authors’ responses see attached response letter). All changes are highlighted in yellow in the manuscript.

In addition, there were changes to the authors’ names and institution. Dr J.K. Harrison changed her name to Burton. Dr T. Welsh is now working at the Research Institute for the Care of Older People in Bath and Dr L. Tanajewski is now also working part-time at the Kozminski University, Warsaw, Poland. These changes have been added to the manuscript (also highlighted in yellow).

Kind regards,

Veronika van der Wardt